Author's response to reviews

Title: Public, Environmental, and Occupational Health Research Activity in Arab Countries: Bibliometric, Citation, and Collaboration Analysis

Authors:

waleed sweileh (waleedsweileh@yahoo.com)
Sa’ed Zyoud (saedzyoud@yahoo.com)
Samah Al-jabi (samahjabi@yahoo.com)
ansam sawalha (ansam@najah.edu)

Version: 3  Date: 30 July 2014

Author's response to reviews: see over
Dear editor,

Thanks a lot for the comments and fruitful suggestions regarding the manuscript entitled

“Public, Environmental, and Occupational Health Research Activity in Arab Countries: Bibliometric, Citation, and Collaboration Analysis”

I revised the manuscript based on comments and suggestions. The revision was based on the following points:

Reviewer: Christine Urquhart

Background and discussion were re-written based on the suggestions of the reviewer. The background was made clearer and discussion was better organized.

Conclusion was re-written and made better

Limitation was made in a separate section

Reviewer: Grisel Zacca-González

Period of the study was well explained and mistakes were corrected.

I added a new table (table 5) to endorse the finding regarding Khartoum University

Falagas reference was corrected

A new reference was added to endorse the statements in the background

Availability of journal category through scimago was mentioned in methodology.